CMSC 735 & MSWE 609

Assignment 4B

In the second part of this assignment, you will now apply the error abstraction pr ocess that was

taught in class.  Use the handout to remind yourself of the process (download a copy from the ftp

site if you have lost your original), and apply it to the fault list you created for the PG

requireme nts in Assignment 4A.  Create a list of the errors that you think are in the document.

Then, return to the PG docume nt and search for additional faults that could have been caused by

each of the errors you identif ied.

You must mail your error list and revised fault list to Ryan (
rjones@cs.umd.edu

) by Tuesday,

Nov. 4 at noon .  You should download the template repor t form 2.2 from the class ftp site and fill

in your faults as described (if you have any proble ms, let Ryan know and he will send an ASCII

version of the file via email).  Keep track of how long you spe nd on error abstraction and,

separately, on re-examining the document for faults; you will need this inf ormation to answer

some of the questions at the bottom of the repor t form (the se que stions are for statistical purposes

only and will not be used in deter mining your grade).

Be sure to keep a copy of the completed form 2.2 file for yourself!   You will need to reuse some

of this infor mation on the next phase of this assignment.